Author's response to reviews

Title: Antibacterial and Antibiotic Resistance Modifying Activity of the Extracts from Allanblackia gabonensis, Combretum molle and Gladiolus quartinianus against Gram-negative Bacteria including Multi-Drug Resistant Phenotypes

Authors:

Aime G Fankam (agfankam@yahoo.fr)
Jules R Kuiate (jkuate@yahoo.fr)
Victor Kuete (kuetevictor@yahoo.fr)

Version: 5 Date: 12 June 2015

Author's response to reviews: see over
Dear editor. thanks so much for sending the comments of the reviewer for this manuscript. We have carefully taken it into consideration in preparing our revision, which has hopefully resulted in an improved, clear and more compelling paper. All changes in the text have been marked in red color in the revised manuscript. We hope you will appreciate it.

Reviewer: Ambrose Okem
Reviewer’s report:

Comments

Please provide the ATCC numbers of the reference strains of the bacteria used in the line 84-87

Answer: As several strains were used for each bacterial species, we provided the reference numbers of all of them

Line 158. deleted 'a'

Answer: done